Author's response to reviews

Title: Silent Cerebral Infarct After Cardiac Catheterization as Detected by Diffusion Weighted Magnetic Resonance Imaging: A Randomized Comparison of Radial and Femoral arterial approaches.

Authors:

Michele Hamon (hamon-mi@chu-caen.fr)
Francesco Burzotta (f.burzotta@eudoramail.com)
Catherine Oppenheim (oppenheim-C@APHP.stanne.fr)
Remy Morello (morello-r@chu-caen.fr)
Fausto Viader (viader-f@chu-caen.fr)
Martial Hamon (hamon-m@chu-caen.fr)

Version: 2 Date: 30 November 2006

Author's response to reviews:

wrong version uploaded!